Author’s response to reviews

Title: Genotype-Informed Estimation of Risk of Coronary Heart Disease Based on Genome-Wide Association Data Linked to the Electronic Medical Record

Authors:

Keyue Ding (ding.keyue@mayo.edu)
Kent R Bailey (baileyk@mayo.edu)
Iftikhar J Kullo (kullo.iftikhar@mayo.edu)

Version: 8 Date: 2 November 2011

Author’s response to reviews: see over
Nov 2nd, 2011

Dear Editorial Team,

We have responded your comments point-by-point below.

Sincerely,

Iftikhar J. Kullo
Professor of Medicine
Division of Cardiovascular Diseases
Mayo Clinic
e-mail: kullo.iftikhar@mayo.edu
phone: 507-284-9049

Response

Major revisions (we require the author to make these changes)

References: Please make sure references are cited in numerical order. We notice references 24, 29 and 32 are not cited within the main text. Please make sure references continue to be cited in numerical order once these references are cited.

We have removed references 24, 29, and 32 in the section of 'References'.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tables: please remove the visible vertical lines from your tables

We have removed the visible vertical lines from tables.

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements

We have corrected the table to avoid merged cell crosses rows.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have checked the manuscript carefully.